Author's response to reviews

Title: Cruciferous vegetable intake is inversely associated with lung cancer risk among smokers: a case-control study

Authors:

Li Tang (Li.Tang@RoswellPark.org)
Gary R Zirpoli (Gary.Zirpoli@RoswellPark.org)
Vijayvel Jayaprakash (Vijay.Jayaprakash@RoswellPark.org)
Mary E Reid (Mary.Reid@RoswellPark.org)
Susan E McCann (Susan.McCann@RoswellPark.org)
Chukwumere E Nwogu (Chumy.Nwogu@RoswellPark.org)
Yuesheng Zhang (Yuesheng.Zhang@RoswellPark.org)
Christine B Ambrosone (Christine.Ambrosone@RoswellPark.org)
Kirsten B Moysich (Kirsten.Moysich@RoswellPark.org)

Version: 5 Date: 20 April 2010

Author's response to reviews:

1. In page 1 author list, delete extra space between GRZ.
2. In Background section, paragraph 1, line 14, delete "-" after NNK.
3. In page 12, line 2, delete extra space before "the".